Author's response to reviews

Title: Gender differentials in the evolution of cigarette smoking habits in a general European adult population from 1993-2003

Authors:

Michael C Costanza (michael.costanza@hcuge.ch)
Julian Salamun (julien.salamun@hcuge.ch)
Alan D Lopez (a.lopez@sph.uq.edu.au)
Alfredo Morabia (a.morabia@hcuge.ch)

Version: 7 Date: 29 April 2006

Author's response to reviews:

Dear BMC,

The following formatting changes (immediately below) were made to the revised manuscript submitted here per your last email (copied below the changes).

Sincerely,
Michael Costanza

************************************************************************************

Formatting changes requested

Qualifications - Please remove the authors' qualifications (e.g. PhD, M.D.) from the title page.
...DONE

Running head Please remove the running head from the title page.
...DONE

Word count Please remove this from the manuscript.
...DONE

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.
...DONE

Website addresses - Please cite website addresses in the reference list and remove them from the manuscript text. Cite them using the following format:
Website title [http://www.website.com].
Where possible, please avoid using very long addresses.
...DONE

References The reference list should contain all authors names, regardless of how many there are. The term et al should not be used.
...DONE

EndNote or Refman references If your references are formatted using EndNote or Refman, please make any changes in the EndNote or Refman record, rather than editing the references in the manuscript manually.
...DONE (references entered manually)

************************************************************************************
Authors: Michael C Costanza, Julian Salamun, Alan D Lopez and Alfredo Morabia
Title: Gender differentials in the evolution of cigarette smoking habits in a general European adult population from 1993-2003
Journal: BMC Public Health
MS: 2489645327773106

Dear Prof Costanza,

We are delighted to confirm that BMC Public Health will publish your manuscript (above) pending formatting requirements. We ask that you make the following changes to the format of your manuscript so that online production process of your manuscript is straightforward. After making these changes, please resubmit a new version of your paper to the submission system. The instructions to submit your revised manuscript are available below.

Formatting changes requested
-------------------------------
Qualifications - Please remove the authors' qualifications (e.g. PhD, M.D.) from the title page.

Running head Please remove the running head from the title page.

Word count Please remove this from the manuscript.

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.

Website addresses - Please cite website addresses in the reference list and remove them from the manuscript text. Cite them using the following format: Website title [http://www.website.com]. Where possible, please avoid using very long addresses.

References The reference list should contain all authors names, regardless of how many there are. The term et al should not be used.

EndNote or Refman references If your references are formatted using EndNote or Refman, please make any changes in the EndNote or Refman record, rather than editing the references in the manuscript manually.

-------------------------------

Please note that this is the proofing stage. All revised manuscripts submitted to BMC Public Health are regarded as being in the final version. You will not be able to make any corrections to your article once it has been accepted and published in preliminary form. Therefore, when you submit your final version, please carefully check your manuscript for any typing or content errors that might have inadvertently occurred, and for small changes to the text that you or your co-authors wish to make at this time. This will be the last chance you have to make any such corrections as any errors present in the manuscript may end up in the final published version. Please submit the revised version of your manuscript via the online submission system: http://www.biomedcentral.com/manuscript/login/man.asp?txt_nav=chk

To submit your revised manuscript
-------------------------------

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted to journals published by BioMed Central.
2. Click on the 'Submit revision' button next to the title of this manuscript to go through to a submission form.
3. With the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version. It is very important that all changes are updated on this page, as well as in the manuscript file as the information on this page will be used in PubMed and on BioMed Central.

4. With the 'Cover letter' tab, please provide a covering letter with a point-by-point description of the changes made.

5. With the 'Upload files' tab, please upload the revised version of the manuscript and press 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments. Please do not hesitate to contact us if you have any problems or questions.

Recent version of the manuscript:
http://www.biomedcentral.com/media/man/2489645327773106/msd/main/0

Figures:
http://www.biomedcentral.com/media/man/2489645327773106/msd/FIG/1
http://www.biomedcentral.com/media/man/2489645327773106/msd/FIG/2

We would ask that you return any revisions by Friday, 05 May 2006.

With all best wishes,

The BioMed Central Editorial Production Team
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: corepreaccept@biomedcentral.com
Web: http://www.biomedcentral.com/